Author's response to reviews

Title: Psychiatric Disorder Criteria and their Application to Research in Different Racial Groups

Authors:

   Nancy C Low (lown@mail.nih.gov)
   John Hardy (hardyj@mail.nih.gov)

Version: Date: 21 December 2006

Author's response to reviews:

To BMC Psychiatry,

The following changes have been made to this accepted CORRESPONDENCE manuscript based
Formatting changes requested below:

----------------------------------------------------------------------
1. Title - Please use sentence case in the title, both in the manuscript file and the submission system (i.e. remove all unnecessary capitals) - CHANGED.

2. Qualifications - Please remove the authors' qualifications (e.g. PhD / MD) from the manuscript and submissions system - REMOVED.

3. Authors' statements - Please remove this from the title page - REMOVED.

4. Competing interests - Please include a competing interests section after the main body of text. Please write 'The authors declare that they have no competing interests' - STATEMENT MADE AND ADDED AT END OF TEXT.

5. Authors' contributions - For the Authors' contributions we suggest the following kind of format before the Acknowledgements (please use initials to refer to each author's contribution): AB carried out the molecular genetic studies, participated in the sequence alignment and drafted the manuscript. JY carried out the immunoassays. MT participated in the sequence alignment. ES participated in the design of the study and performed the statistical analysis. FG conceived of the study, and participated in its design and coordination. All authors read and approved the final manuscript - AUTHORS' CONTRIBUTION STATEMENT ADDED TO END OF TEXT.

6. Keywords section - Please remove the keywords section from the manuscript - REMOVED.

7. Word count - Please remove the word count from the manuscript - REMOVED.

8. Abstract - Please update your submission system abstract so that it is identical to the manuscript - THE ABSTRACT HAS BEEN UPDATED IN THE SYSTEM AND IS IDENTICAL TO THE MANUSCRIPT.

9. Section headings - Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals). For example Abstract not ABSTRACT - CORRECTED.

10. Manuscript sections - Manuscript sections should include (in the following order): Abstract, Background, Results, Discussion, Conclusions, Methods, Abbreviations, Authors' contributions, Acknowledgements, References, Figure legends (if any), Tables (if any), Description of Additional files (if any)- THIS MANUSCRIPT IS A "CORRESPONDENCE" TYPE AND THEREFORE DOESN'T CONTACT ALL OF THESE SECTION TITLES. WE HAVE EXAMINED BMC PSYCHIATRY CORRESPONDENCE EXAMPLES ONLINE AND BELIEVE THIS UPDATED MANUSCRIPT IS IN COMPLIANCE WITH THE SAME FORMAT OF THE TYPICAL CORRESPONDENCE FORMAT. PLEASE LET US KNOW IF THIS IS NOT THE CASE.

11. References - In the manuscript text, please place the reference before the full stop. - CORRECTED

----------------------------------------------------------------------

Thank you and please do not to contact us again for any further changes/revisions.

John Hardy
Nancy Low